import { TypeList } from "docs-ui"

# customer

## Interfaces

- [FindConfig](../../customer/interfaces/customer.FindConfig/page.mdx)
- [CustomerAddressDTO](../../customer/interfaces/customer.CustomerAddressDTO/page.mdx)
- [FilterableCustomerAddressProps](../../customer/interfaces/customer.FilterableCustomerAddressProps/page.mdx)
- [FilterableCustomerGroupProps](../../customer/interfaces/customer.FilterableCustomerGroupProps/page.mdx)
- [FilterableCustomerGroupCustomerProps](../../customer/interfaces/customer.FilterableCustomerGroupCustomerProps/page.mdx)
- [FilterableCustomerProps](../../customer/interfaces/customer.FilterableCustomerProps/page.mdx)
- [CustomerGroupDTO](../../customer/interfaces/customer.CustomerGroupDTO/page.mdx)
- [CustomerGroupCustomerDTO](../../customer/interfaces/customer.CustomerGroupCustomerDTO/page.mdx)
- [CustomerDTO](../../customer/interfaces/customer.CustomerDTO/page.mdx)
- [GroupCustomerPair](../../customer/interfaces/customer.GroupCustomerPair/page.mdx)
- [CreateCustomerAddressDTO](../../customer/interfaces/customer.CreateCustomerAddressDTO/page.mdx)
- [UpdateCustomerAddressDTO](../../customer/interfaces/customer.UpdateCustomerAddressDTO/page.mdx)
- [CreateCustomerDTO](../../customer/interfaces/customer.CreateCustomerDTO/page.mdx)
- [CustomerUpdatableFields](../../customer/interfaces/customer.CustomerUpdatableFields/page.mdx)
- [CustomerGroupUpdatableFields](../../customer/interfaces/customer.CustomerGroupUpdatableFields/page.mdx)
- [CreateCustomerGroupDTO](../../customer/interfaces/customer.CreateCustomerGroupDTO/page.mdx)
- [ICustomerModuleService](../../customer/interfaces/customer.ICustomerModuleService/page.mdx)
- [BaseFilterable](../../customer/interfaces/customer.BaseFilterable/page.mdx)
- [SoftDeleteReturn](../../customer/interfaces/customer.SoftDeleteReturn/page.mdx)
- [RestoreReturn](../../customer/interfaces/customer.RestoreReturn/page.mdx)
- [JoinerServiceConfigAlias](../../customer/interfaces/customer.JoinerServiceConfigAlias/page.mdx)
- [JoinerServiceConfig](../../customer/interfaces/customer.JoinerServiceConfig/page.mdx)
- [IModuleService](../../customer/interfaces/customer.IModuleService/page.mdx)
- [MessageAggregatorFormat](../../customer/interfaces/customer.MessageAggregatorFormat/page.mdx)
- [IMessageAggregator](../../customer/interfaces/customer.IMessageAggregator/page.mdx)
- [Context](../../customer/interfaces/customer.Context/page.mdx)

## Type Aliases

- [MetadataType](../../customer/types/customer.MetadataType/page.mdx)
- [Query](../../customer/types/customer.Query/page.mdx)
- [ExpandScalar](../../customer/types/customer.ExpandScalar/page.mdx)
- [Scalar](../../customer/types/customer.Scalar/page.mdx)
- [ReadonlyPrimary](../../customer/types/customer.ReadonlyPrimary/page.mdx)
- [Primary](../../customer/types/customer.Primary/page.mdx)
- [OperatorMap](../../customer/types/customer.OperatorMap/page.mdx)
- [FilterValue2](../../customer/types/customer.FilterValue2/page.mdx)
- [FilterValue](../../customer/types/customer.FilterValue/page.mdx)
- [PrevLimit](../../customer/types/customer.PrevLimit/page.mdx)
- [FilterQuery](../../customer/types/customer.FilterQuery/page.mdx)
- [EventMetadata](../../customer/types/customer.EventMetadata/page.mdx)
- [MessageBody](../../customer/types/customer.MessageBody/page.mdx)
- [Message](../../customer/types/customer.Message/page.mdx)
- [RawMessageFormat](../../customer/types/customer.RawMessageFormat/page.mdx)
- [JoinerRelationship](../../customer/types/customer.JoinerRelationship/page.mdx)
- [LinkModulesExtraFields](../../customer/types/customer.LinkModulesExtraFields/page.mdx)
- [ModuleJoinerConfig](../../customer/types/customer.ModuleJoinerConfig/page.mdx)
- [ModuleJoinerRelationship](../../customer/types/customer.ModuleJoinerRelationship/page.mdx)
